Author's response to reviews

Title: Risk of uncomplicated peptic ulcer disease in a cohort of new users of low dose acetylsalicylic acid for secondary prevention of cardiovascular events

Authors:

Ana Ruigomez (aruigomez@ceife.es)
Saga Johansson (Saga.Johansson@astrazeneca.com)
Péter Nagy (Peter.Nagy@astrazeneca.com)
Mar Martín-Pérez (mar.martinpe@gmail.com)
Luis A García Rodríguez (lagarcia@ceife.es)

Version: 2  Date: 2 July 2014

Author's response to reviews:

The following has been added to the methods:

'Ethical approval for the study was obtained from a Multicentre Research Ethics Committee (NHS; MREC reference number: 08/H0305/49).'